Author’s response to reviews

Title: The influence of socioeconomic factors on choice of infant male circumcision provider in rural Ghana; a community level population based study

Authors:
Thomas Gyan (thomas.gyan@uwa.edu.au)
Kimberley McAuley (kimberley.mcauley@uwa.edu.au)
Natalie A. Strobel (natalie.strobel@uwa.edu.au)
Sam Newton (samkofinewton@yahoo.com)
Seth Owusu-Agyei (seth.owusu-agyei@kintampo-hrc.org)
Karen Edmond (karen.edmond@uwa.edu.au)

Version: 6 Date: 14 Aug 2017

Author’s response to reviews:
Dear Editor, Thank you once again for the review of our paper. We address the Editor’s comments below, we have made these changes in the revised manuscript and have attached a clean and track change version

BPED-D-16-00715R5
The influence of socioeconomic factors on choice of infant male circumcision provider in rural Ghana; a community level population based study Thomas Gyan, MPH; Kimberley McAuley, PhD; Natalie A. Strobel, PhD; Sam Newton, PhD; Seth Owusu-Agyei, PhD; Karen Edmond, PhD BMC Pediatrics

1. Please ensure that all author names on the title page are in the format: First name, Surname.

• We have now done this

2. Please also ensure that the author details in the system are updated with any changes
• We have made this change in the system.

3. And that the initials used in the Authors’ contributions section reflect this.

• We agree with your suggestion and have done this. Please see page 22, lines 11 to 12

4. Please copy the ‘Role of funding source’ statement in the Methods and include it in the ‘Funding’ section in the Declarations.

• We have done this. Please see the “Funding” sections in the Declarations.

5. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

• We have included this information in the manuscript on page 22.

6. All Tables larger than one A4 in length must be moved to the end of the manuscript, after the References.

• We have done this.